\begin{thebibliography}{10}
\providecommand{\url}[1]{#1}
\def\UrlFont{\rmfamily}
\providecommand{\newblock}{\relax}
\providecommand{\bibinfo}[2]{#2}
\providecommand\BIBentrySTDinterwordspacing{\spaceskip=0pt\relax}
\providecommand\BIBentryALTinterwordstretchfactor{4}
\providecommand\BIBentryALTinterwordspacing{\spaceskip=\fontdimen2\font plus
\BIBentryALTinterwordstretchfactor\fontdimen3\font minus
  \fontdimen4\font\relax}
\providecommand\BIBforeignlanguage[2]{{%
\expandafter\ifx\csname l@#1\endcsname\relax
\typeout{** WARNING: IEEEtran.bst: No hyphenation pattern has been}%
\typeout{** loaded for the language `#1'. Using the pattern for}%
\typeout{** the default language instead.}%
\else
\language=\csname l@#1\endcsname
\fi
#2}}

\bibitem{sas}
``{SAS Institute},'' \url{www.sas.com/}.

\bibitem{BHH:Statistics}
G.~E.~P. Box, W.~G. Hunter, and J.~S. Hunter, \emph{Statistics for
  Experimenters}.\hskip 1em plus 0.5em minus 0.4em\relax New York: John Wiley
  \& Sons, 1978.

\bibitem{CART}
L.~Breiman, J.~Freidman, R.~Olshen, and C.~Stone, \emph{Classification and
  Regression Trees}.\hskip 1em plus 0.5em minus 0.4em\relax Monterey, CA:
  Wadsworth, 1984.

\bibitem{oats}
R.~Brownlie, J.~Prowse, and M.~S. Padke, ``Robust testing of {AT\&T
  PMX}/{S}tar{MAIL} using {OATS},'' \emph{{\em AT}\&{\em T Technical Journal}},
  vol.~71, no.~3, pp. 41--7, 1992.

\bibitem{burr}
K.~Burr and W.~Young, ``Combinatorial test techniques: Table-based automation,
  test generation and code coverage,'' in \emph{Proceedings of the Intl. Conf.
  on Software Testing Analysis \& Review}, 1998.

\bibitem{Childers:03}
B.~Childers, J.~Davidson, and M.~Soffa, ``{Continuous Compilation: A New
  Approach to Aggressive and Adaptive Code Transformation},'' in
  \emph{Proceedings of the International Parallel and Distributed Processing
  Symposium}, Apr. 2003.

\bibitem{dalal}
S.~R. Dalal, A.~Jain, N.~Karunanithi, J.~M. Leaton, C.~M. Lott, G.~C. Patton,
  and B.~M. Horowitz, ``Model-based testing in practice,'' in \emph{Proceedings
  of the Intl. Conf. on Software Engineering, (ICSE)}, 1999, pp. 285--294.

\bibitem{dunietz}
I.~S. Dunietz, W.~K. Ehrlich, B.~D. Szablak, C.~L. Mallows, and A.~Iannino,
  ``Applying design of experiments to software testing,'' in \emph{Proceedings
  of the Intl. Conf. on Software Engineering, (ICSE '97)}, 1997, pp. 205--215.

\bibitem{Grundy:01}
J.~Grundy, Y.~Cai, and A.~Liu, ``{Generation of Distributed System Test-beds
  from High-level Software Architecture Description},'' in \emph{{16 $^th$
  International Conference on Automated Software Engineering, Linz
  Austria}}.\hskip 1em plus 0.5em minus 0.4em\relax IEEE, Sept. 2001.

\bibitem{kola}
W.~Kolarik, \emph{Creating Quality: Systems, Concepts, Strategies and
  Tools}.\hskip 1em plus 0.5em minus 0.4em\relax McGraw-Hill Education, 1995.

\bibitem{Schmidt:05w}
A.~S. Krishna, A.~Gokhale, D.~C. Schmidt, V.~P. Ranganath, J.~H. White, and
  D.~C. Schmidt, ``{Model-driven Middleware Specialization Techniques for
  Software Product-line Architectures in Distributed Real-time and Embedded
  Systems},'' in \emph{Proceedings of the MODELS 2005 workshop on MDD for
  Software Product-lines}, Half Moon Bay, Jamaica, Oct. 2005.

\bibitem{Schmidt:04k}
A.~S. Krishna, D.~C. Schmidt, A.~Porter, A.~Memon, and D.~Sevilla-Ruiz,
  ``{Improving the Quality of Performance-intensive Software via
  Model-integrated Distributed Continuous Quality Assurance},'' in
  \emph{Proceedings of the 8th International Conference on Software
  Reuse}.\hskip 1em plus 0.5em minus 0.4em\relax Madrid, Spain: ACM/IEEE, July
  2004.

\bibitem{Schmidt:04n}
A.~S. Krishna, N.~Wang, B.~Natarajan, A.~Gokhale, D.~C. Schmidt, and G.~Thaker,
  ``{CCMPerf: A Benchmarking Tool for CORBA Component Model Implementations},''
  in \emph{Proceedings of the 10th Real-time Technology and Application
  Symposium (RTAS '04)}.\hskip 1em plus 0.5em minus 0.4em\relax Toronto, CA:
  IEEE, May 2004.

\bibitem{kuhn}
D.~Kuhn and M.~Reilly, ``An investigation of the applicability of design of
  experiments to software testing,'' \emph{Proceedings 27th Annual NASA
  Goddard/IEEE Software Engineering Workshop}, pp. 91--95, 2002.

\bibitem{GME:01}
A.~Ledeczi, A.~Bakay, M.~Maroti, P.~Volgysei, G.~Nordstrom, J.~Sprinkle, and
  G.~Karsai, ``{Composing Domain-Specific Design Environments},'' \emph{IEEE
  Computer}, pp. 44--51, Nov. 2001.

\bibitem{LiuY:02}
Y.~Liu, I.~Gorton, A.~Liu, N.~Jiang, and S.~Chen, ``{Designing a Test Suite for
  Empirically-based Middleware Performance Prediction},'' in \emph{{40 $^{th}$
  International Conference on Technology of Object-Oriented Languages and
  Systems, Sydney Australia}}.\hskip 1em plus 0.5em minus 0.4em\relax
  Australian Computer Society, Aug. 2002.

\bibitem{Lu-FCS:02}
C.~Lu, J.~A. Stankovic, G.~Tao, and S.~H. Son, ``{Feedback Control Real-Time
  Scheduling: Framework, Modeling, and Algorithms},'' \emph{Real-Time Systems
  Journal}, vol.~23, no. 1/2, pp. {85--126}, July 2002.

\bibitem{mandl}
R.~Mandl, ``Orthogonal {L}atin squares: an application of experiment design to
  compiler testing,'' \emph{Communications of the ACM}, vol.~28, no.~10, pp.
  1054--1058, 1985.

\bibitem{Schmidt:04b}
A.~Memon, A.~Porter, C.~Yilmaz, A.~Nagarajan, D.~C. Schmidt, and B.~Natarajan,
  ``{Skoll: Distributed Continuous Quality Assurance},'' in \emph{Proceedings
  of the 26th IEEE/ACM International Conference on Software Engineering}.\hskip
  1em plus 0.5em minus 0.4em\relax Edinburgh, Scotland: IEEE/ACM, May 2004.

\bibitem{Memon01c}
A.~M. Memon, M.~E. Pollack, and M.~L. Soffa, ``Hierarchical gui test case
  generation using automated planning,'' \emph{IEEE Transactions on Software
  Engineering}, vol.~27, no.~2, pp. 144--155, February 2001.

\bibitem{d-optimal}
T.~Mitchell, ``An algorithm for the construction of the 'd-optimal'
  experimental designs,'' \emph{Technometrics}, vol.~16, no.~2, pp. 203--210,
  1974.

\bibitem{sqc96}
D.~C. Montgomery, \emph{Introduction to Statistical Quality Control},
  3rd~ed.\hskip 1em plus 0.5em minus 0.4em\relax New York: John Wiley \& Sons,
  1996.

\bibitem{EMULab:03}
{Robert Ricci and Chris Alfred and Jay Lepreau}, ``{A Solver for the Network
  Testbed Mapping Problem},'' \emph{{SIGCOMM Computer Communications Review}},
  vol.~33, no.~2, pp. 30--44, Apr. 2003.

\bibitem{Schmidt:00a}
D.~C. Schmidt, M.~Stal, H.~Rohnert, and F.~Buschmann, \emph{{Pattern-Oriented
  Software Architecture: Patterns for Concurrent and Networked Objects, Volume
  2}}.\hskip 1em plus 0.5em minus 0.4em\relax New York: Wiley \& Sons, 2000.

\bibitem{tuk77}
J.~W. Tukey, \emph{Exploratory Data Analysis}.\hskip 1em plus 0.5em minus
  0.4em\relax Reading, Massachusetts: Addison-Wesley, 1977.

\bibitem{OCML_ACMSE:04}
E.~Turkay, A.~Gokhale, and B.~Natarajan, ``{Addressing the Middleware
  Configuration Challenges using Model-based Techniques},'' in
  \emph{Proceedings of the 42nd Annual Southeast Conference}.\hskip 1em plus
  0.5em minus 0.4em\relax Huntsville, AL: ACM, Apr. 2004.

\bibitem{White02}
B.~White and J.~L. et~al, ``{An Integrated Experimental Environment for
  Distributed Systems and Networks},'' in \emph{Proceedings of the Fifth
  Symposium on Operating Systems Design and Implementation}.\hskip 1em plus
  0.5em minus 0.4em\relax Boston, MA: USENIX Association, Dec. 2002, pp.
  255--270.

\bibitem{wu}
C.~F.~J. Wu and M.~Hamada, \emph{Experiments: Planning, Analysis, and Parameter
  Design Optimization}.\hskip 1em plus 0.5em minus 0.4em\relax Wiley, 2000.

\bibitem{cohenTSE06}
C.~Yilmaz, M.~B. Cohen, and A.~Porter, ``Covering arrays for efficient fault
  characterization in complex configuration spaces,'' \emph{IEEE Transactions
  on Software Engineering}, vol.~31, no.~1, pp. 20--34, Jan. 2006.

\bibitem{ca}
------, ``Covering arrays for efficient fault characterization in complex
  configuration spaces.'' in \emph{ISSTA}, 2004, pp. 45--54.

\bibitem{ICSE05}
C.~Yilmaz, A.~S. Krishna, A.~Memon, A.~Porter, D.~C. Schmidt, A.~Gokhale, and
  B.~Natarajan, ``Main effects screening: a distributed continuous quality
  assurance process for monitoring performance degradation in evolving software
  systems,'' in \emph{ICSE '05: Proceedings of the 27th international
  conference on Software engineering}.\hskip 1em plus 0.5em minus 0.4em\relax
  New York, NY, USA: ACM Press, 2005, pp. 293--302.

\bibitem{Padua:03}
K.~Yotov, X.~Li, G.~Ren, M.~Cibulskis, G.~DeJong, M.~Garzaran, D.~Padua,
  K.~Pingali, P.~Stodghill, and P.~Wu, ``{A Comparison of Empirical and
  Model-driven Optimization},'' in \emph{{Proceedings of ACM SIGPLAN conference
  on Programming Language Design and Implementation}}, June 2003.

\bibitem{zhang:02}
R.~Zhang, C.~Lu, T.~Abdelzaher, and J.~Stankovic, ``{Controlware: A Middleware
  Architecture for Feedback Control of Software Performance},'' in
  \emph{Proceedings of the International Conference on Distributed Systems
  2002}, July 2002.

\end{thebibliography}

% $Id: skoll_summary.tex,v 1.26 2006/07/24 02:36:28 aporter Exp $
\section {An Overview of the Skoll DCQA Environment} 
\label{skolloverview}

To improve the quality of per\-for\-mance-inten\-sive systems across
large configuration spaces, our work focuses on {\em distributed
continuous quality assurance} (DCQA) processes~\cite{Schmidt:04b} that
evaluate various software qualities, such as portability, performance
characteristics, and functional correctness ``a\-round-the-world and
around-the-clock.''\footnote{Naturally, DCQA processes can also be
executed effectively in more constrained and smaller-scale
environments, such as company-wide intranets.}  To support this
methodology, we developed {\em Skoll}, which is a {\em model-driven
engineering} (MDE)-based DCQA environment
(\url{www.cs.umd.edu/projects/skoll}).  Skoll divides QA processes
into multiple {\it tasks}, each of which is implemented as a generic
process parametrized by one of several alternative configurations
expressed via MDE tools.  Example tasks might include running
regression tests in one of many system configurations, evaluating
system response time under one of several different input workloads,
or measuring code execution coverage using one of several
instrumentation schemes.
\begin{figure}[tbhp]
\centering \epsfig{file=figures/SkollTasksAndSubTasks.eps,scale=0.95}
\caption{The Skoll Distributed Continuous Quality Assurance Architecture} 
\label{fig:dcqa}
\end{figure} 
As shown in Figure~\ref{fig:dcqa}, these tasks are then intelligently
and continuously distributed to -- and executed by -- clients across a
grid of computing resources contributed by end-user and developer
communities. The results of individual tasks are returned to servers
at central collection sites, where they are merged and analyzed to
steer subsequent iterations and ultimately to complete the overall QA
process.

This section summarizes Skoll's key components and services, which
include MDE tools for modeling system configurations and their
constraints, a {\em domain-specific modeling language} (DSML) to
describe these models, algorithms for scheduling and remotely
executing tasks via planning technology that analyzes task results and
adapts the DCQA process in real time, a DSML to package the subtasks,
and techniques to interpret and visualize the results.

\noindent{{\bf QA task space.}}  Performance-intensive systems, such
as the ACE+\-TAO+\-CIAO QoS-enabled middleware, provide a range ({\em
i.e.}, $\sim$500) of configuration options that can be used to tune
its behavior.\footnote{All the ACE+TAO+CIAO's configuration options
are described at
\url{www.cs.wustl.edu/~schmidt/ACE_wrappers/TAO/docs/Options.html}.}
To be effective, DCQA processes must keep track of these options, in
addition to other environmental information, such as OS platform,
build tools used, and desired version numbers.  This information is
used to parameterize generic QA tasks and aids in planning the global
QA process, {\em e.g.}, by adapting the process dynamically and
helping interpret the results.

%
\iffalse
The cornerstone of Skoll is therefore its formal model of a DCQA
process' QA task space.  This model captures different configuration
options, environmental characteristics, their allowable settings, 
and any constraints among these. 
\fi
%

In Skoll, tasks are generic processes parameterized by QA task
options.  These options capture information that is (1) varied under
QA process control or (2) needed by the software to build and execute
properly.  These options are generally application-specific, including
workload parameters, operating system, library implementations,
compiler flags, or run-time optimization controls.  Each option must
take its value from a discrete number of settings.  For example, in
% Done: Adam, Atif, can you please add a citation for this other work?
other work, our QA task model include a configuration option called
{\tt OperatingSystem} so Skoll can select appropriate binaries and
build code for specific tasks~\cite{cohenTSE06}.

\iffalse                 % Commenting it out from here
% Folks, I do not think we use this anywhere else and this probably
% does not add to the paper since our goal is to give a quick overview
% - Andy
Defining a subtask involves mapping each option to one of its
allowable settings.  We call this mapping a {\em configuration} and
represent it as a set $\{$ $(V_1, C_1)$, $(V_2, C_2)$, $\ldots$,
$(V_N, C_N)$ $\}$, where each $V_i$ is an option and $C_i$ is its
value, drawn from the allowable settings of $V_i$.  In practice, not
all configurations are semantically useful, \eg, feature X may not be
supported on operating system Y.  We therefore allow {\it inter-option
constraints} that limit the setting of one option based on the setting
of another.

Skoll represents constraints as ($P_i \rightarrow P_j$), meaning ``if
predicate $P_i$ evaluates to $TRUE$, then predicate $P_j$ must
evaluate to $TRUE$.''  A predicate $P_k$ can be of the form $A$, $\neg
A$, $A \& B$, $A | B$, or simply ($V_i = C_i$), where $A$, $B$ are
predicates, $V_i$ is an option and $C_i$ is one of its allowable
values.  A {\em valid configuration} is a configuration that violates
no inter-option constraints.
\fi                      % Commenting it out to here

\noindent{{\bf QA task modeling.}}  The QA task model underlies the
DCQA process.  Our experience~\cite{Schmidt:04b,Schmidt:04k,ca} with
the initial Skoll prototype taught us that building these models
manually was tedious and error-prone.  We therefore developed and
integrated into Skoll the {\em Options Configuration Modeling Language
(OCML)}~\cite{OCML_ACMSE:04}. OCML is an MDE tool that provides a DSML
for modeling software configurations.  For example, OCML defines a
{\em numeric option} type for middleware options that can have numeric
values, {\em e.g.}, cache, buffer, or thread pool sizes.  OCML is
built atop the {\em Generic Modeling Environment (GME)}~\cite{GME:01},
which provides a meta-programmable framework for creating DSMLs and
generative tools via {\em metamodels} and {\em model interpreters}.
For the feasibility study in Section~\ref{experiment}, we used the
OCML MDE tool to define the configuration model visually and generate
the low-level formats used by other Skoll components.

\iffalse                 % Commenting it out from here
% probably don't need these details. - andy
We applied OCML to model and capture the various configuration
parameters provided by ACE+\-TAO+\-CIAO.  Figure~\ref{fig:ocml} shows
the TAO ORB configuration options modeled with OCML.
\begin{figure}[htpb]
\centerline{\epsfxsize=3in,\epsffile{figures/rule2.eps}}
\caption{OCML Modeling Paradigm Applied to TAO ORB}
\label{fig:ocml}
\end{figure}
Inter-option dependencies are captured as model constraints. For
instance, Figure~\ref{fig:ocml} shows that when the {\tt
ORBConnectionPurgingStrategy} is set to value {\tt NULL}, then neither
{\tt ORB\-Con\-nectionCacheMax} nor {\tt
ORBConnectionCachePurgePercentage} can be declared or given a
user-specified value.  OCML validates the modeled middleware configurations
and generates configuration files that can be used to configure the
middleware.  
For the feasibility study in Section~\ref{experiment}, we
used the OCML MDE tool to visually define the configuration model and
generate the low-level formats used by other Skoll components.
\fi                      % Commenting it out to here

\noindent{{\bf Exploring the QA task space.}} Since the QA task spaces
of many systems can be enormous, Skoll contains an {\em Intelligent
Steering Agent} (ISA) \cite{Schmidt:04b} that uses AI planning
techniques~\cite{Memon01c} to distribute QA tasks on available Skoll
clients.  When clients become available they send a message to the
Skoll server.  Skoll's ISA then decides which task to assign it by
considering many factors, including (1) {\em the QA task model}, which
characterizes the subtasks that can be assigned legally, (2) {\em the
results of previous tasks}, which capture what tasks have already been
done and whether the results were successful, (3) {\em global process
goals}, such as testing popular configurations more than rarely used
ones or testing recently changed features more heavily than unchanged
features, and (4) {\em client characteristics and preferences}, {\em
e.g.}, the selected configuration must be compatible with the OS
running on the client machine or users can specify preferences that
configurations must run with user-level -- rather than superuser-level
-- protection modes.

After a valid configuration is chosen, the ISA packages the
corresponding QA task into a {\em job configuration}, which consists
of the code artifacts, configuration parameters, build instructions,
and QA-specific code ({\em e.g.}, developer-supplied
regression/\-performance tests) associated with a software pro\-ject.
Each job configuration is then sent to a Skoll client, which executes
the job configuration and returns the results to the ISA.  By default,
the ISA simply stores these results.

In some experiments, however, we want to learn from incoming results.
For example, when some configurations prove faulty, it makes no sense
to retest them. Instead, we should refocus resources on other
unexplored parts of the QA task space.  When such dynamic behavior is
desired, DCQA process designers develop customized {\it adaptation
strategies} that Skoll uses to monitor the global process state,
analyze it, and use the information to modify future task assignments
in ways that improve process performance.  One example of an
adaptation strategy is the {\it nearest neighbor search strategy},
which allows a process to target failing QA task subspaces by
preferentially testing the ``neighbors'' of a failing configuration,
({\em i.e.}, other similar configurations that differ in one
configuration option value) to see if they also
fail~\cite{Schmidt:04b}.

\noindent{{\bf Packaging QA tasks.}} With the initial Skoll prototype,
developers who wanted to evaluate QoS issues had to provide
hand-written benchmark programs.  For example, ACE+TAO+CIAO developers
creating such benchmarks to measure latency and throughput for a
particular workload had to write (1) the header files and source code
that implement the measurements, (2) the configuration and script
files that tune the underlying ORB and automate running tests and
output generation, and (3) project build files ({\em e.g.}, makefiles)
required to generate executable binaries from source code.  Our
initial feasibility study~\cite{Schmidt:04b} revealed that this
process was tedious and error-prone.

To address these problems, we developed the {\em Benchmark Generation
Modeling Language (BGML)}~\cite{Schmidt:04n}, which is an MDE tool
that automates key QoS evaluation concerns of QoS-enabled middleware
and applications, such as (1) modeling how distributed system
components interact with each other and (2) representing metrics that
can be applied to specific configuration options and platforms.
Middleware/application developers can use BGML to graphically model
interaction scenarios of interest.  BGML automates the task of writing
repetitive source code to perform benchmark experiments and generates
syntactically and semantically valid source and benchmarking code.

\iffalse                 % Commenting it out from here
To model interactions, BGML provides (1) {\em test elements}, such as
operations, return-types, latency and throughput, that can be used to
represent generic operations or a sequence of operation steps and
associate functional QoS properties with them and (2) {\em workload
elements}, such as tasks and task-set that could be used to model and
simulate background load present during the experimentation process.
BGML maps these workload elements to individual platform-specific code
in the interpretation process.
\fi                      % Commenting it out to here

\begin{figure}[htpb]
\centerline{\epsfxsize=5.1in\epsffile{figures/BGML-Operation.eps}}
\caption{Associating QoS with an Operation in BGML} 
\label{bgml-operation}
\end{figure}
Figure~\ref{bgml-operation} depicts how QA engineers can visually
configure an experiment that measures end-to-end latency.  As shown in
the figure, the latency metric was associated with an operation ({\tt
WorkOrderResponse()}) using BGML.  BGML's {\tt TaskSet} element was
also used to create background tasks that invoked the {\tt
WorkOrderResponse()} operation continuously for a fixed number of
iterations.

\noindent{{\bf Analysis of results.}} Since DCQA processes can be
complex, Skoll users often need help to visualize, interpret, and
leverage process results.  Skoll therefore supports a variety of
pluggable analysis tools, such as Classification Tree Analysis
(CTA)~\cite{CART}.  In previous work~\cite{Schmidt:04b,ca}, we used
CTA to diagnose options and settings that were the likely causes of
specific test failures.  For the work presented in this paper, we
developed statistical tools to analyze data generated by the
formally-designed experiments described next.
